Author's response to reviews

Title: Child maltreatment and hypertension among young adults

Authors:

Shakira F Suglia (sfs2150@cumc.columbia.edu)
Cari J Clark (cjclark@umn.edu)
Renee Boynton-Jarrett (Renee.BoyntonJarrett@bmc.org)
Nancy R Kressin (nkressin@bu.edu)
Karestan C Koenen (kck5@columbia.edu)

Version: 5 Date: 22 September 2014

Author's response to reviews: see over
Dear Prof Latkin,

We very much appreciate the continued interest in our manuscript. As requested we have responded to the reviewers comments and provide details of our point-by-point response below, showing the reviewers comments in italics followed by our response in normal font.

1. Provide Conclusion section.

We have named the last paragraph of our discussion section, our conclusion section as it summarizes our findings and provides an explanation of the paper’s importance and relevance.

2. Rename conflict of interest to competing interest

*We have made the change.*